Author's response to reviews

Title: Epidemiology of nausea and vomiting of pregnancy: prevalence, severity, determinants, and the importance of race/ethnicity.

Authors:

Anaïs Lacasse (anais.lacasse.morris@umontreal.ca)
Évelyne Rey (Evelyne_Rey@SSSS.gouv.qc.ca)
Ema Ferreira (ema.ferreira@umontreal.ca)
Caroline Morin (caroline.morin.hsj@ssss.gouv.qc.ca)
Anick Berard (anick.berard@umontreal.ca)

Version: 4 Date: 26 June 2009

Author's response to reviews: see over
June 26th, 2009

Dr. Melissa Norton, MD
Editor-in-Chief
BMC Pregnancy and Childbirth
BioMed Central Ltd
Floor 6, 236 Gray’s Inn Road
London, UK
WC1X 8HL

Dear Editor:

We thank you for reviewing our manuscript entitled “Epidemiology of nausea and vomiting of pregnancy: prevalence, severity, determinants, and the importance of race/ethnicity.” (MS: 5321668712492008)

We modified the manuscript according to the formatting requirements made by the editorial board. The description of the modifications made to our manuscript is detailed below.

Formatting changes

- As suggested, accurate details were provided for reference # 38.
- Table 3 & 4 were resubmitted as two additional files. Therefore, these tables were removed from the manuscript, a section entitled “Additional files” was added to the end of the manuscript, and additional files (1 & 2) were mentioned in the text.
- Unnecessary capitals were removed from the title.
- All “track changes” were removed from the manuscript.
- Keywords were also removed.
- We verified that all the text was black.
- All vertical lines were removed from the tables.
- Figure 1 was cropped as closely as possible to minimise white space around the image.

All authors have read and approved the revised version of the manuscript. The formatted manuscript is now ready for publication in BMC Pregnancy and Childbirth.

Sincerely,

Anaïs Lacasse, PhD
Research Center, CHU Sainte-Justine
3175, Côte-Sainte-Catherine
Montreal (Quebec) H3T 1C5
Phone: (514) 345-4931 ext 4272
Fax: (514) 345-4801
E-mail: anais.lacasse.morris@umontreal.ca